Title: Metaplastic ossification in the cartilage of the bronchus of a chronic multi-drug resistant tuberculosis patient: a case report

Authors:

Seok-Yong Eum (syeumkr@gmail.com)
Ji-Hye Kong (jhkong@itrc.re.kr)
Bo-Young Jeon (bojeon87@gmail.com)
Sang-Nae Cho (raycho@yonsei.ac.kr)
Jhingook Kim (jkimsmc@skku.edu)
Laura E Via (lvia@niaid.nih.gov)
Clifton E Barry III (cbarry@niaid.nih.gov)
Won-Jung Koh (wjkoh@skku.edu)

Version: 4 Date: 9 February 2010

Author's response to reviews: see over
Editor
“Journal of Medical Case Reports”

9 February, 2010

Dear Editor,

This cover letter accompanies our submission of the revised version of MS ID 6688014212383746 entitled “Metaplastic ossification in the cartilage of the bronchus of a chronic multi-drug resistant tuberculosis patient”. This is the most recent version of the manuscript and we hope that you find the revised manuscript suitable for publication in the Journal.

Sincerely,

Seok-Yong Eum, DVM, Ph.D.
Division of Immunopathology and Cellular Immunology